NBS Working paper 9/2023

Leaning against housing booms fueled by credit

Carlos Cañizares Martínez



www.nbs.sk



© Národná banka Slovenska 2023 research@nbs.sk

This publication is available on the NBS website www.nbs.sk/en/publications-issued-by-the-nbs/research-publications

The views and results presented in this paper are those of the authors and do not necessarily represent the official opinion of the National Bank of Slovakia.

ISSN 2585-9269 (online)

Leaning against housing booms fueled by credit*

Carlos Cañizares Martínez[†]

October 5, 2023

Abstract

This study aims to empirically identify the state of the US housing market and establish a countercyclical state-dependent macroprudential policy rule. I do so by estimating a Markov switching model of housing prices, in which mortgage debt affects house prices nonlinearly and drives state transition probabilities. Second, I propose a state-contingent policy rule fed with the probability of being in each state, which I apply to setting a housing countercyclical capital buffer, a mortgage interest deduction, and a dividend payout restriction. Finally, I show that such hypothetical tools contain early warning information in a forecasting exercise to predict the charge-off rates of real estate residential loans and a financial stress index. The significance of this study is that it informs policymakers about the state of the housing market mechanically, while also providing a general rule to implement a state-contingent and timely macroprudential policy.

Keywords: House prices, non-linear modeling, Markov switching model, household debt, macroprudential policy.

JEL Classification: C22, C24, G51, R21, R31.

^{*}This paper is an improved version of the first chapter of my PhD in Economics at the University of Milano-Bicocca which was previously circulated as "Housing booms fueled by credit". I would like to thank Matteo Pelagatti and Vittoria Cerasi for their advice and Andrea Bastianin for his constructive comments. Three anonymous referees, John Duca, Dirk Bezemer, Jan Klacso, Adriana Lojschova, Vaclav Zdarek, Joan Paredes, Alessandra Donini and other seminar participants at the "Minsky at 100" conference at the Università Cattolica del Sacro Cuore (Milan, 2019), at the 28th Annual Symposium of the Society for Nonlinear Dynamics and Econometrics (virtual, 2020), at the RCEA-Europe Conference on Global Threats to the World Economy (virtual, 2023), at internal seminars at Banco de Portugal and the National Bank of Slovakia, at the 27th International Conference on Macroeconomic Analysis and International Finance (Rethymno, 2023), at the 12th International Conference of the Financial Engineering and Banking Society (Chania, 2023), and at the 3rd Catalan Economic Society Conference (Barcelona, 2023) provided me interesting discussions and suggestions.

[†]Department of Economic and Monetary Analysis, *National Bank of Slovakia*, *Imricha Karvaša* 1, 811 07 Bratislava, Slovakia. Email: carlos.canizares.martinez@nbs.sk.

1 Introduction

The Global Financial Crisis and related literature on the effects of housing and credit booms in the economy raised a consensus about the danger that such phenomena pose to financial stability. Indeed, a large body of evidence identifies excessive credit growth as a good predictor of financial crises (e.g., Cerutti et al., 2017b). A direct consequence of this realized turmoil was the introduction of new banking regulations to avoid the excessive build-up of credit risk. However, setting a macroprudential policy to mitigate housing boom-bust cycles, which might need to be nonlinear to achieve better economic stabilization (see Gatt, 2023), is still an open issue. In particular, the current methods for estimating housing price overvaluation and excessive credit do not seem fit for the job. Current approaches to monitoring housing booms often do not include mortgage debt in their models, neglecting a key reason for systemically dangerous housing overvaluation (Crowe et al., 2013; Jordà et al., 2015a). Additionally, the appearance of new policy tools typically carries uncertainty about how to implement them. Such threats might provide weak comfort to policy-makers in implementing timely countercyclical policies, in turn giving room for *inaction bias*. In addition, electoral cycles may further dampen fiscal policymakers' incentives to actively deal with housing booms fueled by credit (Müller, 2022).

In the last decade, theoretical studies incorporating financial frictions and rich mortgage markets in macro models have investigated the mortgage credit channel of transmission (see Favilukis et al., 2017; Justiniano et al., 2019; Greenwald, 2018; and Guerrieri and Iacoviello, 2017, among others). New modeling devices such as collateral and lending constraints introduced in these models allow the generation of strong housing prices and credit boom-bust cycles that are consistent with those observed in the US and other countries during the 2000s. The nonlinearities inherent in these models around whether borrowing constraints are binding prescribe the usage of a nonlinear macroprudential policy (Guerrieri and Iacoviello, 2017; Gatt, 2023). However, before taking any action, policymakers must identify the current housing market state.

This study addresses this question by estimating a Markov switching model in which housing price growth is explained by standard housing demand fundamentals plus mortgage debt growth, which affects house prices nonlinearly and also drives the transition probabilities between each state. In doing so, I focus on identifying periods of *housing booms fueled by credit*, that is, states in which there is a high growth in house prices and mortgage debt once controlling for standard housing demand fundamentals¹. In this way, I narrow down the search for this systemic risk source along the lines of Crowe et al., (2013). Thus, I distinguish between housing market states that allow for countercyclical and state-dependent macroprudential policies, as suggested theoretically by Gatt (2023) and empirically by Drehmann et al. (2010).

The described approach overcomes some key challenges faced by alternative methods for monitoring housing price development. First, the literature most often tries to find bubbles or explosive behavior in house prices, which is not necessarily a threat to financial stability because what makes a bubble systemically dangerous is being financed with debt (Jordà et al., 2015a). For this reason, bubble tests are not well-suited for macroprudential policy discussions. Second, the models used

¹Alternatively, housing bubbles are usually defined along the lines of Stiglitz (1990), who establishes that "if the reason that the price is high today is only because investors believe that the selling price is high tomorrow—when 'fundamental' factors do not seem to justify such a price—then a bubble exists". Therefore, a bubble is fueled by beliefs or sentiments, whereas it does not make any explicit assumptions about the role of credit.

for this purpose are very stylized, leaving room for relevant variable omissions. Third, econometric models used for identifying overvaluation often provide inconsistent results (ESRB, 2022).

The contributions of this study are as follows. First, I provide a new empirical tool, namely, a Markov switching model, to identify the state of the housing market, focusing on identifying *housing booms fueled by credit* instead of asset price bubbles, thereby providing a stronger narrative for triggering a macroprudential policy. Second, I exploit the probabilities of being in each housing market state obtained in the regime-switching model to establish a rule for setting a sectoral countercyclical capital buffer (SCCyB)², mortgage interest deduction, and dividend payout restriction. Interestingly, such a policy rule structure is sufficiently general to be applied to other policy tools. Finally, I show that the results obtained by the proposed regime-switching model and policy rule contain useful early warning information about the soundness of banks' housing loans and financial risk in a forecasting exercise.

Related literature. This paper is related to several strands of the literature. First, this study's approach is motivated by empirical studies analyzing the effects of housing booms accompanied by credit booms on the economy and financial stability (Mian and Sufi, 2009, 2010, 2011, 2018, Jordà et al., 2013, 2015a, 2015b, 2016, 2017, Cerutti et al. 2017b, Greenwood et al. 2022, and Schularick and Taylor, 2012, among others). These studies document that credit booms tend to boost housing prices and, in turn, increase the risk of financial crises. Therefore, it is crucial for policymakers to monitor the housing market by focusing on identifying housing booms fueled by credit.

Second, this study is related to theoretical studies that model credit-driven housing booms using macro models. While the introduction of housing sectors in macro models arrived only in the 2000s, in recent years, additional features such as collateral and lending constraints have allowed modelers to generate realistic housing boom-bust cycles (see Favilukis et al., 2017 and Justiniano et al., 2019). Moreover, Guerrieri and Iacoviello (2017) find asymmetric effects in the relationship between house prices and economic activity owing to an asymmetry in the effects of collateral constraints. More recently, Gatt (2023) shows that occasionally binding borrowing constraints are a source of nonlinearity that warrants a nonlinear macroprudential policy. I follow this literature to motivate using a regime-switching model of housing booms fueled by credit.

Third, this analysis is also complementary to empirical studies measuring or dating periods of house price overvaluation, where there are three different approaches. One approach consists in estimating a "fundamental" house price using an econometric model, and deriving the deviations from actual prices, an approach that is extensively used in central banks (ESRB, 2022). A second avenue is testing for mildly explosive behavior, as Phillips et al. (2011, 2015) propose, which exploits augmented Dickey-Fuller tests to find *exuberance* in asset prices. A third and relatively underused approach is to employ regime-switching models to date periods characterized by high or low prices (Van Norden and Schaller, 1993, 1996). Instead, this study focuses on housing booms fueled by credit, which I argue is better suited for setting macroprudential policy. Therefore, this study also relates to the household red-zone of Greenwood et al. (2022), which is an indicator of contemporaneous high household credit and housing price growth.

²As defined by the BIS (2019a), a sectoral CCyB is an additional capital requirement that "would require banks to build up a capital buffer on exposures to credit segments in which credit developments are deemed excessive."

Finally, this study also relates to the literature on macroprudential policies that deal with creditdriven housing booms. During the 2000s, there was no consensus on the appropriate policy response to housing booms (Bernanke, 2002; Roubini, 2006). The acute turmoil amid the Global Financial Crisis made it clear that better banking regulation would help mitigate systemic risks, namely via the so-called borrower-based measures (Aikman, 2021) and capital buffers (BIS, 2011, 2019a, 2019b). Despite being available to regulators for almost a decade, calibrating them remains challenging. For example, the countercyclical capital buffer (CCyB), in which standard suggestions on how to set it (BIS, 2010) do not seem to be widely followed (Döme and Sigmund, 2023).

The remainder of this paper is organized as follows. Section 2 presents the theoretical framework and the Markov switching model of housing booms fueled by credit. Section 3 describes the data and model specification and presents the empirical results. Section 4 proposes a general policy rule for setting a housing SCCyB, mortgage interest deduction, and dividend payout restriction by exploiting the estimated probabilities of each state using the Markov switching model. Section 5 assesses the early warning content of the estimated hypothetical macroprudential tools by evaluating their ability to predict banking losses in residential real estate loans and a proxy for financial risk. Section 6 discusses the significance of the results and the limitations of this study. Finally, Section 7 concludes.

2 Modeling housing booms fueled by credit

2.1 Theories of housing booms fueled by credit

Both empirical and theoretical research has highlighted the critical roles of financial accelerator effects and house prices in explaining the boom and bust of the 2000s that triggered the Great Recession. This subsection explores prominent theoretical frameworks that have emerged to elucidate the interactions among credit expansion, housing markets, and macroeconomic outcomes. The studies presented here serve as cornerstones for understanding the interplay of factors that contribute to the emergence and amplification of credit-fueled housing booms without being extensive, which motivates and builds the theoretical foundations of the Markov switching model of housing booms fueled by credit presented in this section.

The seminal paper in this literature, Iacoviello (2005), embeds mortgage debt and house prices into a New Keynesian DSGE model. This study integrates nominal household debt and a collateral constraint (Kiyotaki and Moore, 1997) tied to real estate prices into the DSGE model. The study reveals that demand shocks influence housing and consumer prices in the same direction, thereby amplifying them. The rise in asset prices enhances debtors' borrowing capacity, allowing them to spend more and invest more. An increase in consumer prices reduces the real value of outstanding debt obligations, which positively affects their net worth. Since borrowers have a higher propensity to spend than lenders, the net effect on demand is positive.

Building on this foundation, several studies have used this framework to analyze several features of the US boom and bust of the 2000s³. Favilukis et al. (2017) use a quantitative general equilibrium model with housing and collateral constraints to explore what drives fluctuations in

³See Guerrieri and Uhlig (2016) for a comprehensive survey of the literature.

house prices to rent ratio, introducing two new features not previously considered: aggregate business cycle risk and bequest heterogeneity in preferences. In this framework, the authors find that a relaxation of collateral requirements can generate a large boom in housing prices, while lower interest rates cannot explain housing booms. They also show that the key mechanism to generate a house price boom is a decline in housing risk premium, namely, the expected future housing return in excess of the interest rate.

To analyze the drivers of the US boom and bust in credit and housing prices that precipitated the Great Recession, Justiniano et al. (2019) use a DSGE model of household borrowing that features a lending constraint, a device to model the expansion in credit supply. The interaction between this constraint and the standard borrowing limit generates rich patterns of debt and home values that significantly improve the model's ability to match several fundamental facts observed during the 2000s' boom. In this way, they focus on looser lending constraints as drivers of the credit booms, consistent with the empirical works of Mian and Sufi (2009), Favara and Imbs (2012), and Di Maggio and Kermani (2014).

Similarly, Greenwald (2018) investigates the role of the *mortgage credit channel* of macroeconomic transmission in a DSGE model with loan-to-value and payment-to-income constraints. A novel propagation mechanism, namely the constraint switching effect, translates into large movements in house prices. This mechanism is active when there are changes in which the two constraints are binding for borrowers. Building on this framework, Greenwald and Guren (2021) analyze the role of credit in driving the 2000s housing boom using a DSGE model with arbitrary intermediate levels of rental markets frictions, thus avoiding taking an extreme modeling choice on the degree of housing supply elasticity. In such a framework, they find that in the US, between one-third and half of the increase in price-rent ratios during the 2000s was due to an increase in credit supply.

Guerrieri and Iacoviello (2017) also explore the relationship between house prices and economic activity in a nonlinear DSGE model, finding an asymmetry between them. They find that financial frictions in collateral constraints matter disproportionately more in a recession than in a boom, leading to asymmetric effects of housing booms and busts depending on whether housing collateral constraints are binding. This mechanism is fundamental for explaining the depth of the Great Recession.

More recently, Gatt (2023) shows that occasionally binding borrowing constraints are a source of nonlinearity that warrants a nonlinear macroprudential policy. Using a DSGE model, this author finds that an asymmetric macroprudential policy rule that lowers the borrowing limit more aggressively during credit booms obtains better economic outcomes than an optimized symmetric rule. Such an asymmetric policy response reduces output and inflation tail risks, generating better economic stabilization and positive externalities to monetary policy.

2.2 A Markov switching model of housing booms fueled by credit

The Markov switching model of house prices adopted in this study is an extension of Markov chains with time-varying transition probabilities, drawing from Hamilton (1989), applied in the housing sector. This model aims to identify different states in the housing sector, with a special emphasis on identifying housing booms fueled by credit. Many empirical (Mian and Sufi, 2009;

Cerutti et al., 2017b; Crowe et al., 2013) and theoretical studies (Guerrieri and Iacoviello, 2017; Justiniano et al., 2019; Greenwald, 2018) underline that this phenomenon plays a crucial role in explaining large boom-bust cycles and thus has strong implications for financial stability; therefore, policymakers must monitor their likelihood. Economic variables such as house prices and mortgage debt growth are a series that undergo episodes in which they seem to change dramatically, suggesting that a regime-switching approach to model housing booms fueled by credit is a promising venue. To that end, this section proposes a Markov switching model of housing booms fueled by credit that provides the probability of being in different housing states.

Let us define HP_t as the log-difference of the nationwide house price index in period t, X_t as a vector of variables whose effects on HP_t are state-independent, Y_t is a vector of variables whose effects on the dependent variable are state-dependent, and s_t is the latent state variable that defines the state of the housing sector in each period t such that:

$$HP_t = \phi_{0,s_t} + \phi_1' X_t + \phi_{2,s_t}' Y_t + \varepsilon_t \tag{1}$$

where $\varepsilon_t \stackrel{\text{iid}}{\sim} N(0, h_{s_t})$. For simplicity, let us assume that there are two states, denoted as 1 and 2, such that $s_t = 1$ or $s_t = 2$. Therefore, depending on the state, the state-dependent coefficients can be either $(\phi_{0,1}, \phi_{2,1}, h_1)$ or $(\phi_{0,2}, \phi_{2,2}, h_2)$.

The state transition probabilities are assumed to follow a first-order Markov chain such that:

$$p_t = P(s_t = 1 | s_{t-1} = 1, \omega_t)$$
(2)

$$1 - p_t = P(s_t = 2 | s_{t-1} = 1, \omega_t)$$
(3)

$$q_t = P(s_t = 2 | s_{t-1} = 2, \omega_t)$$
(4)

$$1 - q_t = P(s_t = 1 | s_{t-1} = 2, \omega_t)$$
(5)

where ω_t is a vector of variables known in period *t* that affect the state transition probabilities in period *t*. Therefore, I am assuming that state transition probabilities are not constant, but time varying and dependent on a vector of economic variables. The parameters of this model are obtained by maximum likelihood estimation. See Appendix B for details on the estimation of the model.

3 Empirical results

3.1 Data

The sample size starts in January 1984, that is, avoiding possible issues arising from the structural break in aggregate volatility before the *Great Moderation*, and ends in March 2023. The house price index used in the baseline Markov switching model is the S&P Case-Shiller home price index, where the alternatives are the house price index computed by the Federal Housing Finance Agency (FHFA, henceforth), and the 10-City and 20-City composites also offered by S&P Case-Shiller. The reasons for using the S&P Case-Shiller home price index as the baseline US housing price time series are twofold. First, it is a nationwide measure that fit the scope of this study. Second, because the data source they use for computing the index relies on the records that are registered in local government deeds recording offices (see S&P Dow Jones Indices, 2019) instead of records in

a particular banking institution, which would make the index a function of the decision making of such firm at different levels such as the lending standards, refinancing and securitization policies⁴.

Beyond housing prices, the main time series used in this study are the fundamental drivers of housing demand and those related to housing finance. As part of the first block, I consider a measure of employment (*all employees: total non-farm payrolls*), wages (*gross domestic income: compensation of employees, paid: wages and salaries*), and housing rental prices (*CPI for urban consumers: rent of primary residence*), which are standard measures of income and purchasing capacity commonly used in the literature. In the second group of variables, I consider a measure of mortgage debt *(mortgage debt outstanding, individuals, and other holders*), which is interpolated to monthly frequency using Chow and Lin (1971) linear interpolation.

Additional variables are used for companion tasks, such as measures of housing demand (*working-age population*) and housing supply (*new one-family houses sold, new building permits*, and *hous-ing starts*). Table 7 in the Appendix lists all of the time series used in this study, their sources, and the transformations performed.

3.2 Model specification

The Markov switching model of housing booms fueled by credit proposed in this study is a monthly model that features house price growth as the dependent variable and four explanatory variables, all of which are specified in quarterly growth rates. In particular, HP_t is the S&P Case-Shiller home price index in month t, whereas the independent variables are measures of wages (gross domestic income: compensation of employees, paid: wages and salaries, denoted by W), employment (all employees: total non-farm payrolls, denoted by E), and housing rental prices (CPI for urban consumers: rent of primary residence, denoted by R), which are fundamental variables of housing demand assumed to affect house prices growth homogeneously across states. Instead, mortgage debt outstanding (specifically, mortgage debt outstanding, individuals, and other holders, denoted by D) is a variable that affects housing prices nonlinearly, depending on the housing market state in each period⁵. Therefore, the Markov switching model is expressed as follows:

$$HP_t = \beta_{0,s_t} + \beta_1 W_t + \beta_2 E_t + \beta_3 R_t + \beta_{4,s_t} D_t + \varepsilon_t$$
(6)

where $\varepsilon_t \sim N(0, h_{st})$. I consider two specifications, Model 1 and Model 2, which are three- and four-state specifications such that $s_t = [1, 2, 3]$ and $s_t = [1, 2, 3, 4]$, respectively. The choice to determine the number of states depends on two elements. First, it is assumed that a reasonable guess in modeling house price dynamics would suggest setting the minimum number of states equal to three, which may correspond to normal times, booms, and bursts. Second, statistical tests showed that adding a fourth state provides a better fit to the data (see subsection 3.3).

Further, the conditional variance of HP_t is given by:

$$\ln(h_{s_t}) = \lambda_{0,s_t} \tag{7}$$

⁴The house price index computed by the Federal Housing Finance Agency relies on records obtained by reviewing repeat mortgage transactions on single-family properties whose mortgages have been purchased or securitized by Fannie Mae or Freddie Mac (see www.fhfa.gov).

⁵It is standard in the literature to employ measures of income and rental prices as fundamental variables of housing prices, and also assuming that overvaluation might be related to non-linear relationships between prices and some determinants, for instance credit (see IMF, 2019 and Gürkaynak, 2008, among others).

The state-transition probabilities for the four-states specification (Model 2) are specified as follows:

$$p_t = P(s_t = 1 \mid s_{t-1} = 1, \omega_t) = \Phi(\pi_{0,p} + \pi_{1,p}D_t)$$
(8)

$$q_t = P(s_t = 2 \mid s_{t-1} = 2, \omega_t) = \Phi(\pi_{0,q} + \pi_{1,q}D_t)$$
(9)

$$z_t = P(s_t = 3 \mid s_{t-1} = 3, \omega_t) = \Phi(\pi_{0,z} + \pi_{1,z}D_t)$$
(10)

$$r_t = P(s_t = 4 \mid s_{t-1} = 4, \omega_t) = \Phi(\pi_{0,r} + \pi_{1,r}D_t)$$
(11)

where p_t , q_t , z_t and r_t are the probabilities of staying in the same state in period t, conditional on staying in the same state in the previous period and the information set ω_t , for states one, two, three, and four respectively. Thus, the transition probabilities across housing market states are timevarying and depend on mortgage debt growth, consistent with the target of identifying housing booms fueled by credit. In this way it is captured the notion of Mian and Sufi (2009), Di Maggio and Kermani (2017), Justiniano et al. (2019), Greenwald (2018), and Stein (2021), among others (see previous section), that credit can induce the appearance of large housing boom-bust cycles⁶. Implicitly, I am making two assumptions in this specification. First, the fundamental dynamics of house prices are well approximated by the standard fundamentals of housing demand and *dividends* such as wages, employment, and rent, which do not have a role in boom-bust state-dependencies. Second, the possible excess demand leading to boom-bust cycles is driven by mortgage debt.

3.3 Results

Table 1 shows the estimated coefficients of Models 1 and 2, that is, the three- and four-state Markov switching models specified in the previous subsection, respectively⁷. The following results stand out. First, state-dependent parameters are most often highly statistically significant. Second, the state-dependent coefficients of mortgage debt outstanding growth are positive, suggesting that an increase in this variable has a positive effect on housing price growth, independent of the state. Third, the constants in state 2 (Model 1) and states 2 and 3 (Model 2) are the only ones with positive coefficients, indicating that they may be booming states. Fourth, the constants in states 3 (Model 1) and 4 (Model 2) are negative and far more negative than those in state 1, which suggests that they may correspond to a burst state. Fifth, the coefficients of wages and rent are positive and highly statistically significant.

⁶The relationship between mortgage debt and house prices may be bidirectional, as already shown in the empirical literature on housing. However, because the target in this model is to capture a state with both high housing prices and debt, disentangling possible reverse causality is not considered necessary.

⁷This Markov switching model is estimated using the toolkit of Ding (2023). To initialize the filter, naïve values for the coefficients are chosen as follows: β_1 , β_2 , β_3 and β_5 are set equal to 0.5; $\beta_{0,s}$ and $\beta_{4,s}$ are set equal to zero; the diagonal transition probabilities in Model 1 (2) are set equal to 0.95 (0.925), and the off-diagonal transition probabilities are set equal to 0.025 (0.025).

	Model 1		Model 2		
Mean parameters					
Constant, state 1	-0.000	(0.00)	-0.004***	(0.00)	
Constant, state 2	0.021***	(0.00)	0.025***	(0.00)	
Constant, state 3	-0.019***	(0.00)	0.006***	(0.00)	
Constant, state 4	-	-	-0.023***	(0.00)	
Wages (W)	0.062**	(0.03)	0.134***	(0.02)	
Employment (E)	0.013	(0.03)	-0.032*	(0.02)	
Rent (R)	0.471***	(0.05)	0.157***	(0.04)	
Mortgage debt (D), state 1	0.537***	(0.03)	0.617***	(0.02)	
Mortgage debt (D), state 2	0.172*	(0.10)	0.064	(0.09)	
Mortgage debt (D), state 3	0.498***	(0.08)	0.405***	(0.02)	
Mortgage debt (D), state 4	-	-	0.753***	(0.11)	
Log likelihood value	1705.	78	1798.	1798.16	
Akaike Information Criterion	-3363.	57	-3518.32		
Bayesian Information Criterion	ı -3263.85 -33		-3356.	28	
Number of estimated parameters	24		39		
Number of observations	471		471		
Number of states	3		4		

Table 1: Markov switching model estimates.

Notes: Standard deviations between brackets. Significance levels at 1%, 5% and 10% are represented by ***, **, * asterisks. Standard errors calculated using the first partial derivatives of the log likelihood, i.e. the outer product matrix. Variance parameters are not reported to save space, while they are all close to zero. Time-varying transition probabilities are reported in Appendix C.

A battery of statistical tests evaluates how the modeling choices in building the specified models fit the data, as summarized in Table 8 in the Appendix. First, a likelihood ratio test for the existence of three states versus two states rejects the null hypothesis in favor of the unrestricted model; in this case, a three-state model specification (Model 1). Furthermore, a likelihood ratio test for the existence of four states versus three states also rejects the null hypothesis in favor of the unrestricted model, thus supporting a four-state specification (Model 2). Second, likelihood ratio tests assessing whether the state-dependent coefficients are equal reject the null hypothesis of equal means, both regarding the constant (β_{0,s_t}) and the coefficient of mortgage debt (β_{4,s_t}) in equation (6), both in Models 1 and 2. A likelihood ratio test of equal variances across states (λ_{0,s_t}) also rejects the null hypothesis of equal variances for both models. The time-varying coefficient of mortgage debt growth on the probability of being in each state ($\pi_{1,p}$, $\pi_{1,q}$, $\pi_{1,z}$, $\pi_{1,r}$) is also tested using a likelihood ratio test, in which the null of equal effects across states is rejected for both models. Finally, a Goldfeld-Quandt heteroskedasticity test focusing on mortgage debt fails to reject the null hypothesis of homoskedasticity in both Models 1 and 2, thus suggesting that the errors are not related to mortgage debt growth.

Figure 1 plots the smoothed (solid red line) and filtered (dashed red line) Markov switching probabilities of being in each state according to Model 2. As previously mentioned and upheld later in this subsection, I argue that states 2 and 3 may correspond to booming states, while state 4

might correspond to *implosion*⁸. Notably, three booming episodes were identified in State 3. First, during the 2000s, from late 2003 to March 2006, that is, the years prior to the Great Recession. Second, there was a boom between April 2012 and April 2014. Finally, there was also a booming signal early after the Covid-19 pandemic hit, from August 2020 to June 2022. These three episodes appear to occur in times of highest real house price growth (black line) and match the probabilities of being in state 2 according to Model 1 (see Figure 4 in the Appendix). Times in which the probabilities of being in state 2 are high also correspond to episodes in which house prices and real mortgage debt outstanding growth (purple line) are booming, though at a slower speed than in state 3. Interestingly, the probability of being in state 2 was close to one during the early 2000s, from 2000 to 2004, hinting at early booming signals. In addition, it was high during the mid- 1980s and between 2014 and 2018. In contrast, the highest probabilities of being in state 4 correspond to episodes in which house prices declined heavily, such as in the early 1990s, the late 2000s during the Great Recession, and the post-pandemic slump in 2022. Alternatively, the high probabilities of being in state 1 coincide with years in which house price growth is close to zero and during the recovery of 2009–2012.

⁸A specific and commonly agreed upon definition of housing booms and credit booms is missing in the literature, where such empirical definitions are typically ad-hoc. For instance, Crowe et al. (2013) define a real estate boom as a period in which real house price appreciation is above a threshold of 1.5% or the annual real house price appreciation rate exceeds the country-specific historical annual appreciation rate. They also define a credit boom as a period in which the growth rate of bank credit to the private sector in % of GDP is more than 20 % or exceeds the rate implied by a country-specific, backward-looking, cubic time trend by more than one standard deviation.



Figure 1: Markov switching probabilities of being in state s_t , Model 2.

Table 2 presents the latest transition probability matrix for Model 2^9 . This shows that once in any of the four states, the most likely subsequent outcome is remaining in the same state. Moreover, when in state 2, the most likely alternative possibility is to move to state 3, which means moving from a booming state to another more exuberant state in terms of house price growth, as shown in Table 4. Furthermore, in implosion state 4, the only alternative to continue in the same state is to move to state 1, which is consistent with the idea that after an implosion, it is unlikely to transition immediately to a booming state again.

		State in T-1			
		1	2	3	4
	1	0.96	0.00	0.02	0.03
State in T	2	0.01	0.93	0.04	0.00
	3	0.02	0.07	0.91	0.00
	4	0.01	0.00	0.03	0.97

Table 2: Latest transition probabilities matrix, Model 2.

Table 3 presents the expected duration in each state for the two estimated models (see Hamilton, 1989). To calculate this, I assume that the probability of being in each state S is the average of the latest transition probabilities of being in state S estimated in each period t across all the time periods.

	Model 1	Model 2
State 1	45.21	27.89
State 2	19.64	14.96
State 3	22.12	24.81
State 4	-	15.14

Table 3: Expected duration in each state.

Notes: Expected duration is expressed in the number of time periods, which is the expected number of months.

To further investigate the nature of the estimated housing market states, Table 4 shows the average growth rates of some macroeconomic and financial variables depending on the state in which the US housing market is estimated based on the results of Model 2 (except interest rates, which are on average levels). These calculations correspond to the Markov switching model results with smoothed probabilities, whereas the differences with respect to the filtered estimation are negligible. The following observations emerge from these results. First, real house price growth is positive and much higher than for the full sample average during states 2 and 3, while it is strongly negative in state 4 and close to zero during state 1. This suggests that states 2 and 3 correspond to housing price booms, whereas state 4 corresponds to bursts, and state 1 is a steady state. Second, it turns out that standard housing demand fundamentals develop in a relatively smoother fashion, thus suggesting that large swings in house prices should be driven by other forces. Third, mortgage

⁹Additionally, the latest transition probabilities matrix of Model 1 is shown in Appendix E.

debt grows at the highest speed during state 2, whereas it also grows at the full sample average or more during states 3 and 4. Moreover, mortgage rates and Fed funds are relatively lower in states 2 and 3. Finally, real estate indicators exhibit expansionary behavior during the booming states 2 and 3 and a markedly negative behavior amid *implosion* state 4, with declines in housing starts, new building permits, houses sold, new homes under construction, cement production, and accumulation of house supply. Other macro variables, such as industrial production, sales, and unemployment levels, also exhibit a contractionary tone during state 4. Overall, these descriptive statistics are consistent with interpreting state 1 as a steady state that includes normal and recovery times, state 2 and 3 as *housing booms fueled by credit*, and state 4 as *implosion* times. Regarding the difference between states 2 and 3, the latter exhibits a house price growth exuberance that seems beyond the development of both standard housing demand fundamentals and credit.

	Full Sample	State 1	State 2	State 3	State 4
Housing prices	A				
Real S&P Case-Shiller home price index	0.14	-0.03	0.39	0.83	-0.49
Real urban primary residence rent index	0.05	-0.01	0.13	-0.05	0.12
Housing demand fundamentals					
Non-farm employees	0.11	0.16	0.06	0.22	0.04
Real wages	0.17	0.22	0.20	0.14	0.04
Working age population (aged 15-64)	0.07	0.07	0.08	0.05	0.07
Financial variables					
Real mortgages debt outstanding	0.27	0.13	0.40	0.26	0.32
30-year mortgage fixed rate average*	6.76	7.74	5.75	4.36	8.41
Fed funds*	3.54	4.45	2.59	0.73	5.56
Real real estate loans securitized	0.16	0.48	0.22	-0.70	-0.02
Other macroeconomic variables					
Industrial production	0.14	0.33	0.05	0.27	-0.17
Real manufacturing and trade sales	0.20	0.35	0.18	0.23	-0.09
Real estate indicators					
Housing starts	0.26	0.48	0.60	1.24	-1.57
New building permits	0.12	0.43	0.65	0.71	-1.99
New one family houses sold	0.24	0.78	0.53	0.13	-1.39
Supply of houses	0.47	-0.49	0.55	2.14	1.08
New homes under construction	0.04	-0.11	0.58	1.31	-1.39
Real cement production	0.06	0.29	0.31	0.02	-0.90

 Table 4: Summary statistics in each state: growth averages (%).

Notes: Average growth rates computed for each subset of data, except in the case of interest rates, in which they are the average of the variable in levels. Full sample ranges from January 1984 to March 2023. The quarterly variables were linearly interpolated to obtain a monthly series.

3.4 Comparison with alternative overvaluation signals

In this subsection, I focus on booming states 2 and 3 and compare the results with those of alternative approaches. To this end, I first construct measures of housing price overvaluation using two dynamic common factor models to proxy for housing demand and supply overvaluation, which may enrich the interpretation of overvaluation sources. The structure of the dynamic factor models is common for both specifications and is specified in detail in Appendix F. In summary, I take a few indicators of demand and supply and summarize them separately using a dynamic common factor model following an autoregressive structure of order two. Then, I define overvaluation from demand (supply) as present when the house price growth rate exceeds that of the common factor of demand (supply).

Second, I compare my results with those provided by researchers using mildly explosive behavior (MEB) tests to identify bubbles in house prices along the lines of Phillips et al. (2011, 2015). In the literature, I mostly focus on Shi (2017) and Fabozzi et al., (2020), given that they do not rely on a price-fundamental ratio to test for exuberance but instead test the residual of an estimation using several fundamentals. However, a key caveat must be highlighted: these authors try to find housing bubbles, while in this paper, I define a model to find housing booms fueled by credit. In other words, Figure 2 plots the probabilities of being in a *housing boom fueled by credit* (state 2 in graph A; state 3 in graphs B, C, and D), together with the overvaluation signals (i.e., the gray areas) according to the demand (graphs A and B) and supply side (graph C) dynamic common factor models, and the mildly explosive behavior test results of Shi (2017) in the bottom graph (brown area).

During the years prior to the Global Financial Crisis in the 2000s, Model 2 points to a high probability of being in state 2 from 2000 to 2004 and high probabilities of being in state 3 from 2004 to 2006. During this time span, overvaluation signals point to the role of demand factors in pushing such a housing boom. These results are consistent with the literature, which widely interprets the housing boom of the 2000s as a result of booming demand fostered by credit (see Duca et al., 2010; Favara and Imbs, 2015; Di Maggio and Kermani, 2017; Adelino et al., 2018; and *inter alia*). The dating of such housing boom fueled by credit period is along the lines of the findings of some researchers testing for housing bubbles using both user cost econometric models (Muellbauer, 2012), mildly explosive behavior tests (Shi, 2017; Fabozzi et al., 2020; Coulter et al., 2022; Pavlidis et al., 2016), and regime-switching techniques (Nneji et al., 2013; Whitehouse et al., 2023). Interestingly, Shi (2017) identified a housing bubble from 2004 to 2005, which coincided with the final two years of this *housing boom fueled by credit*. Moreover, Greenwood et al. (2022) also find a household red-zone between 2002 and 2006 in the US, which coincides with high probabilities of being in states 2 and 3 in the Markov switching Model 2, while the latter started providing booming signals in 2000.

More recently, a *housing boom fueled by credit* emerged from March 2012 to April 2014. In this period, there were signals of overvaluation from demand and supply, whereas the explosiveness test of Shi (2017) was silent. Indeed, an inelastic housing supply providing less than demanded housing is a common reading of this period (Rappaport, 2016; JCHS Harvard, 2018, among others).

Finally, a *housing boom fueled by credit* emerged in August 2020 after the pandemic hit, which lasted until mid-2022, when real house price growth increased significantly. This booming signal is contemporaneous with both the demand and supply overvaluation signals. Notably, this housing boom signal appeared before the exuberance indicators employing MEB tests on house price fundamentals provided overvaluation evidence (Coulter et al., 2022).



Notes: Probabilities of being in each state using Model 2 results. DCFM means dynamic common factor model, which provides house prices overvaluation signals plotted in gray (see Appendix F for a description). MEB tests are the mildly explosive behavior tests as reported by Shi (2017) and the Dallas Fed (see https://www.dallasfed. org/research/international/houseprice#tab2).

4 Countercyclical macroprudential policy

One of the critical challenges faced by central banks is the management of credit-driven booms and busts, characterized by alternating periods of exuberant credit expansion and abrupt contractions, often leading to systemic risks and economic vulnerabilities. Motivated by the desirability of asymmetric macroprudential policy (Gatt, 2023, Cerutti et al., 2017a), this section proposes a state-dependent macroprudential policy rule that exploits the previously defined nonlinear model of housing booms fueled by credit, which is expressed as follows:

$$Y_{t} = \phi_{1} \cdot Y_{t-1} + \phi_{2} \cdot \left[\sum_{i=1}^{4} P(S_{t} = i)_{t} \cdot c_{i}\right]$$
(12)

where Y is the level of a particular policy tool chosen at time t, ϕ_1 and ϕ_2 are smoothing parameters that sum up to one, $P(S_t = i)_t$ are the probabilities of being in states $S_t = [1,2,3,4]$ as estimated in each month t by the Markov switching Model 2 shown in previous sections¹⁰. Notably, the coefficients c_i where i = [1,2,3,4] are the policy tool target levels chosen depending on the housing market state¹¹. While coefficients c_i must be chosen for each policy tool and state, the smoothing parameters ϕ_1 and ϕ_2 are assumed to be always set equal to 0.9 and 0.1, respectively.

The motivation for using equation (12) is as follows. First, policymakers need to know the state of the credit and housing markets as a precondition for implementing a countercyclical policy aimed at taming credit-driven booms and busts. This is informed by the state probabilities $P(S_t = i)_t$ estimated using the Markov switching model. Second, after choosing a certain policy tool *Y*, an active policymaker must decide which policy levels to impose in each state, corresponding to the parameters c_i . Importantly, these parameters must be selected such that they are countercyclical. Third, the smoothing parameters ϕ_1 and ϕ_2 govern the speed at which the policy level in *t* adjusts from the level chosen in t - 1, thus allowing the policymaker to determine the smoothness of the desired policy tool. Fourth, Equation (12) is general, such that one can use a different model to inform the state of the economy, and it can also be used to estimate the levels of diverse macroprudential policy tools, as shown below.

To show how to set a countercyclical macroprudential policy using equation (12), I apply this policy rule to set three hypothetical tools: a sectoral countercyclical capital buffer, a mortgage interest deduction, and a dividend payout restriction. Although these tools are available to policymakers, they appear relatively underused in most jurisdictions, including the US, despite their potential to deal with real estate booms (Crowe et al., 2013). First, the sectoral countercyclical capital buffer (SCCyB) was proposed by the BIS (2019a, 2019b), which is a variation of the traditional countercyclical capital buffer and is an additional capital requirement for banks during periods of excessive credit growth concentrated within specific sectors, such as real estate, construction, or consumer credit. Furthermore, the literature shows that higher capital requirements for banks are an effective way to increase the resilience of the banking sector and contain both credit and housing prices (see Ampudia et al., 2021 and Mendicino et al., 2020, among others). Using a sectoral

¹⁰Therefore, Y_t is conditional on the information that feed the Markov switching Model 2, that is: S&P Case-Shiller home price index, measures of employment (all employees: total non-farm payrolls), wages (gross domestic income: compensation of employees, paid: wages and salaries), housing rental prices (CPI for urban consumers: rent of primary residence), and mortgage debt (mortgage debt outstanding, individuals, and other holders).

¹¹The state-dependent policy target levels for each policy tool discussed in this section were chosen for expositional purposes. In a more realistic exercise, they may be calibrated and/or time-varying.

version of the tool such that the focus is on risks associated with the housing sector seems to be a natural choice for this application.

The second policy tool considered in this application is the mortgage interest rate deduction (MID) at the disposal of governments. This tax provision allows individuals who own a home and have a mortgage to deduct a portion of the interest they pay on their mortgage loan from their taxable income. In many countries, including the US, this deduction is typically provided as an incentive to promote homeownership at the cost of stimulating the housing market. Over the last two decades, several studies have discussed the effects of this tool since Gervais (2002). Despite the disagreement in the literature with respect to the short-run costs of a transition to deduction removal, there appears to be consensus on the long-run welfare gains associated with the elimination of this deduction (see Karlman et al., 2021). Moreover, given that this deduction has a direct impact on government revenue, a countercyclical approach to this tool could help governments obtain a fiscal space that might be useful during implosion times to support financial stability. Thus, this policy might allow governments to incorporate financial stability considerations into the design of fiscal policy, as advocated by Borio et al. (2023), while disincentivizing the accumulation of mortgage debt by households during housing booms fueled by credit. Recent theoretical work by Arce et al. (2023) suggests that an optimal tax on borrowing is always desirable macroprudential policy.

In this application, the third policy tool is dividend payout restrictions on banks (DPR). This refers to the imposition of a cap on the maximum dividend banks can deliver to their shareholders, which is a tool for the disposal of central bankers. The rationale for this instrument is that banks could jeopardize their capitalization if they provide sizable dividends during bad times, which might subsequently affect their capacity to provide credit and, if necessary, absorb losses. As noted in Stein (2021), a payout curtail during the Great Recession could have attenuated the impact of the ongoing banking crisis on the real economy. Furthermore, Acharya et al. (2017) show in their theoretical framework that the interconnections between banks' capital policies can generate strong externalities, leading to excessive dividends and inefficient capitalization. Along these lines, dividend restrictions were introduced in the US during the Covid-19 crisis, as in many other jurisdictions (see Hardy, 2021). Theoretical work by Muñoz (2021) shows that this tool is effective in smoothing the financial cycle and generates significant welfare gains with respect to traditional capital regulation. In addition, Dautovic et al. (2023) show that the DPR imposed in the Euro area during the pandemic was effective in limiting pro-cyclicality while adding capital space to banks.

To compute the levels of each of the three hypothetical policy tools for each period from Equation (12), I specify the parameters c_i as described in Table 5. The general reasoning behind this parameterization is to choose an expansionary coefficient c_4 to ease credit conditions during busts, instead fixing contractionary parameters c_2 and c_3 during housing booms fueled by credit and considering an intermediate value c_1 during the state 1 times, such that the particular tool is countercyclical. Regarding the SCCyB, I fix the parameters such that $c_1 = 1\%$ (i.e. a positive cycle-neutral SCCyB), $c_2 = 2.5\%$, $c_3 = 4\%$ and $c_4 = 0\%$ in percent of sectoral risk weighted assets (RWA), thus matching the lower and upper bounds suggested by the BIS (2019b), assuming the case of a specialized bank and a contemporaneous broad CCyB requirement equal to 0%. Thus, the banking system is expected to react by limiting households' credit growth during housing booms and easing access to credit during implosion. Instead, the parameterization of the MID is set such that in state 1, households can deduct half of their mortgage debt service costs ($c_1 = 50\%$); however, during the booming state 2 only half of that deduction is possible ($c_2 = 25\%$), and no deduction is possible during the most exuberant housing booms ($c_3 = 0\%$), while households can deduct all their mortgage debt costs during busts ($c_4 = 100\%$). Therefore, potential homeowners are disincentivised to take a mortgage during booms, while they receive a tax incentive to get one during busts. Finally, regarding DPR, the employed parameterization assumes that banks can redistribute 25% of their net earnings during normal times ($c_1 = 25\%$), which is around the actual average after the Great Recession. Instead, they are allowed to distribute up to half of their net earnings during housing booms ($c_2 = 37.5\%$; $c_3 = 50\%$). By contrast, they could not pay any dividend during implosion times ($c_4 = 0\%$), which would guarantee the protection of their capital when competition and signaling might encourage them to pay a risky dividend.

Table 5: Parameterization of the policy tools levels, by state.					
Policy tools	Parameters				
I only tools	c_1	c_2	c_3	С4	
Model 2 (four states)					
Sectoral CCyB (SCCyB)	1%	2.5%	4%	0%	
Mortgage interest rate deduction (MID)	50%	25%	0%	100%	
Dividend payout restriction (DPR)	25%	37.5%	50%	0%	

Notes: Parameters c_1 , c_2 , c_3 , and c_4 correspond to the chosen policy tools levels during state 1, 2, 3, and 4, respectively, for a four-state Markov switching model (Model 2, see Section 2).

Figure 3 plots the levels of the hypothetical macroprudential policies (black solid lines) examined in this section, together with the charge-off rates of real estate loans ("all loans" in blue bars, "residential loans" in green bars) and the probability of being in a housing boom fueled by credit as represented by state 3 probabilities (red solid line) estimated by the Markov switching Model 2 described in previous sections. The charge-off rate of real estate loans is the percentage of banks' debt outstanding that is delinquent or bad debt. Several observations emerge. First, the patterns of the three policy tools are the same or inverted because they are fed by the same probabilities of being in each of the four states provided by the Markov switching model. Second, during the housing boom fueled by credit in the 2000s, the three policy tool levels jumped smoothly towards the chosen contractionary c_3 . As it turns out, periods of high charge-off rates on real estate loans (i.e., during the early 90s and the late 2000s) are preceded by years in which the policy rules prescribe contractionary levels, suggesting leading and countercyclical behavior for such hypothetical policy tools. Furthermore, during the Great Recession, when the charge-off rates of real estate loans reached a maximum, the policy levels hit expansionary c_4 , consistent with the countercyclical reasoning of the selected parameterization.



Figure 3: Hypothetical countercyclical macroprudential policies. **A)** Sectoral countercyclical capital buffer (SCCyB).

Sources: Own calculations and Board of Governors of the Federal Reserve System data. *Notes*: CORREL refers to charge-off rates of real estate loans. Charge-off rates on real estate residential loans is only available from February 1992. In subfigure A, the left hand side scale represents the percent with respect to sectoral RWA (hypothetical SCCyB) or to total loans (CORREL). In subfigure B, the left hand side has 2 scales. LHS 1 (left) reflects the percent of charge-off rates on loans with respect to the total loans of the banking system (CORREL), while the LHS 2 (right) represents the percent of the interest rate paid by borrowers that is deductible (hypothetical MID). In subfigure C, the left hand side scale has 2 scales. LHS 1 (left) reflects the percent of charge-off rates on loans with respect to the total loans of the

banking system (CORREL), while the LHS 2 (right) represents the percent of the dividend payout restriction (hypothetical DPR).

5 Assessment of early warning signaling content

In this section, I delve into an assessment of the early warning content of the hypothetical policy tools described in the previous section, which in turn builds on the probabilities of being in normal times, housing booms fueled by credit, and implosion times, as computed using a Markov switching model. The reasoning behind this exercise is that if those probabilities capture the risks of housing booms and busts, and if the parameterization of the policy rules is countercyclical, such hypothetical policy rules should provide predictive value in forecasting materialized risks in housing finance. This is similar in spirit to Greenwood et al. (2022), who find that their red zone indicators have predictive power for financial crises in a large set of countries.

While the ultimate materialization of macro-financial risk is arguably a financial crisis, a natural leading indicator of housing finance risk is the variable charge-off rate of real estate residential loans, namely, the share of bad housing loans in banks' portfolios. An additional risk indicator considered in this section is the St. Louis Fed Financial Stress Index, which measures the degree of financial stress in the markets and builds mostly on interest rate series and yield spreads. In particular, I implement a forecasting exercise to examine whether the computed hypothetical policy tools contain useful information to predict such risk measures, which would suggest that they are useful in providing early warning signals of real estate booms and financial risks, which, in turn, might suggest that they might be timely in policy-making.

Despite the importance of the riskiness of loan portfolios in the banking system for financial stability, only a few studies tackle the problem of modeling and forecasting the charge-off rate of banking loans, and there is no consensus on which approach to employ¹². Thus, given the presence of modeling uncertainty, a natural approach is to use agnostic models, such as factor models, which allow for dimensionality reduction and to feed them with a large set of macroeconomic predictors. In this exercise, I follow the approach of Kim and Son (2023) and use a factor-augmented vector autoregression (FAVAR) model and a wide US macro dataset, such as the FRED-MD, as a benchmark forecasting model. As these authors show, this type of model outperforms alternative models in predicting the charge of rates of real estate loans.

The forecasting exercise performed in this section is as follows. First, I summarize the monthly FRED-MD dataset, which includes 127 macro and financial US variables based on seven factors, as determined by using the test of Bai and Ng (2002). As previously mentioned, the target variables to forecast are the charge-off rate of real estate residential loans (CORREL), for which the data series starts in February 1992, and the St. Louis Fed Financial Stress Index (STLFSI4), which starts in January 1994. Second, I consider three augmented models, which consist of the benchmark FAVAR model plus one additional policy tool at a time, that is, the hypothetical SCCyB, MID, and DPR, as shown in the previous section. I also report the performance of a naïve model, that is, an autoregressive AR(12) model, including 12 lags. Finally, I evaluate the forecasting performance of the five models up to 12 months ahead considering the subsample January 2007 - February 2020 as the evaluation sample, to avoid distortions coming from the policies implemented to contain the effects of the Covid-19 pandemic. Table 6 reports the relative root mean squared errors (RMSE)

¹²Kim and Son (2023) show that factor models perform well in predicting the charge-off rates of business and real estate loans using a large panel of macro indicators. Barth et al. (2020) find that Ridge regressions, elastic nets and factor models provide accurate forecasts of charge-off rates for four US banks. Sheng et al. (2022) employ a panel Tobit model to assess the importance of uncertainty in forecasting charge-off rates of small US commercial banks.

of the five forecasting models considered with respect to the FAVAR benchmark. As it turns out, the inclusion of hypothetical policy tools increases the forecasting performance of the benchmark FAVAR model to predict CORREL up to 17%, and up to 4% to forecast STLFSI4. These results suggest that despite using a competitive model as a benchmark, the policy tools additionally contain useful information to predict housing finance and financial stress risks, meaning that if implemented, these tools might lead the materialization of risks in the banking sector coming from housing loans and financial stress risks.

tools: Relative RMSE vs FAVAR model.				
Target	CORREL	STLFSI4		
Benchmarks				
FAVAR	1.00	1.00		
AR (12)	1.05	1.16		
Augmented models				
FAVAR with SCCyB	0.93	0.97		
FAVAR with MID	0.83	0.96		
FAVAR with DPR	0.83	0.96		

Table 6: Forecasting performar	nce of the hypothetical
--------------------------------	-------------------------

Notes: Relative RMSE is the ratio of the RMSE of each considered model with respect to the benchmark FAVAR model, which is a monthly FAVAR model with 7 factors summarizing the FRED-MD dataset, Minnesota priors and 11 lags. The AR(12) model is an autorregressive model of order 12. CORREL are the charge-off rates on residential real estate loans of the banking sector. STLFSI4 is the St. Louis Fed Financial Stress Index.

6 Discussion

The outcomes of this research provide insights into a new model to identify the state of the housing market, focusing on catching housing booms fueled by credit and how to exploit it in a statedependent policy rule. Moreover, I show that this procedure can produce policy tools or indicators that provide early warning signaling content to predict proxies for financial risk. In this section, I discuss the differences of the proposed approach with respect to other ones, some policy implications, as well as some of the limitations of this study.

Compared with other approaches to monitoring the housing market, the approach proposed in this study offers the following advantages. First, in contrast to empirical studies targeting the identification of housing bubbles, such as mildly explosive behavior tests, this study builds a regimeswitching model to identify housing booms fueled by credit. I believe that this crucial phenomenon deserves a tailored modeling approach, despite the importance of bubbles, which are slightly different events. Notably, monitoring housing booms fueled by credit might give macroprudential policymakers a stronger narrative for setting housing policies. Second, compared to the household red-zone of Greenwood et al. (2022), the approach taken in this study allows data to speak without imposing specific thresholds to determine risky levels, which might be somewhat ad-hoc and country-dependent. Notwithstanding, a suite-of-models approach to monitoring the housing market welcomes the input from all these complementary approaches. Some policy implications emerge from this study's empirical results. First, during the 2000s housing boom, Markov switching Model 2 started providing booming signals from 1999 (states 2 and 3), which in retrospect might have been useful to policymakers in driving policy tools that, according to the state-dependent policy tools shown in this study, would have been increasingly tight from 1999 to 2006. Second, after the pandemic hit in March 2020, the model continued to provide booming signals that were especially strong from September 2020 to May 2022, thus prescribing policy-tightening. Therefore, according to these results, the easing of policies prevalent at that time might have been unnecessary, at least with regard to the housing market. Nevertheless, it must be noted that these considerations are formulated with the benefit of hindsight.

Regarding the limitations of this study, those related to the Markov switching model are as follows. First, this model does not provide an estimate of the size of overvaluation or undervaluation of house prices, as it is not the target of this study. Second, given the lack of data on foreign housing demand, which is common to most (if not all) economies, such measures were not included in the models. Regarding the proposed state-dependent and countercyclical policy rules, the following caveats emerge. First, the chosen target policy levels are time-invariant and hypothetical. Interested policymakers may prefer to calibrate such targets. Second, the same applies to smoothing parameters, which might be time-varying, such that they require a higher or lower speed of adjustment depending on additional data or the policymaker's judgment.

7 Concluding remarks

Empirical evidence related to excessive leverage, housing price growth, and financial instability has been accumulated since the Global Financial Crisis. As a result, new banking regulations have emerged to reduce the likelihood of housing booms causing macroeconomic damage. The theory suggests that policymakers might succeed in doing so; however, they need at least two ingredients for an effective recipe. First, they must know the housing market state at each point in time. Second, they require a state-contingent policy rule to implement an appropriate countercyclical policy on a timely basis. To date, there has been no consensus among researchers and policymakers regarding how to address both issues.

In this paper, I address such questions using a regime-switching approach. First, I use a Markov switching model to estimate the probability of being in different housing states. Second, I exploit such probabilities to feed a state-contingent policy rule, which I apply to setting a hypothetical housing countercyclical capital buffer (SCCyB), mortgage interest deduction, and dividend payout restriction. Finally, I show that such tools contain early warning information, as they improve the forecasting accuracy of a benchmark model in predicting charge-off rates for real estate residential loans and the St. Louis Fed Financial Stress Index. These results suggest that the three-step approach employed in this study might be useful for housing analysts and policymakers to monitor the housing market and implement state-dependent macroprudential policies in a timely manner.

Future research along these lines may proceed in the following directions. First, it is necessary to perform an impact analysis of the hypothetical macroprudential policies proposed in this study, which may require the use of a non-linear general equilibrium model. Second, the policy rule employed in this study may also be used to drive other policy tools. Finally, the work done in this study may be replicated with data from other countries and also US state-level data.

Appendices

A Data

			D . 4
Variable	Acronym	Source	Data
	•/		transformation
S&P Case-Shiller home price index	HP	S&P	R, L, D
Compensation of employees	W	BEA	R, L, D
Employees, non-farm payrolls	E	BLS	L, D
Urban primary residence rent index	R	BLS	R, L, D
Mortgage debt outstanding, all holders	D	BG	SA, M, R, D
Working age population: aged 15-64	-	OECD	L, D
30 year fixed rate mortgage average	-	FM	-
Fed funds	-	BSL	-
Real estate loans owned and securitized	-	BG	R
Industrial production	-	BG	L
Real manufacturing and trade sales	-	BSL	L
Housing starts	-	CB	L
New private housing building permits	-	CB	L
New one family houses sold	-	CB	L
Supply of new houses	-	CB	L
New private homes under construction	-	CB	L
Industrial production: Cement	-	BG	L, R
Charge-off rate on loans secured by real estate: all loans	CORREL, all	BG	М
Charge-off rate on loans secured by real estate: residential loans	CORREL, resid.	BG	М
St. Louis Fed financial stress index	STLFSI4	BSL	-

 Table 7: Time series data.

Notes: In the last column, L means logs, D means taking one difference, R means that the variable has been transformed into real terms by applying the CPI, and M means that the series has been transformed to a monthly frequency by linear interpolation. Regarding the sources of the data, S&P means Standard & Poor's, BLS means Bureau of Labor Statistics, BEA means Bureau of Economic Analysis, CB means the Census Bureau, BG stands for Board of Governors of the Federal Reserve System, and BSL means Federal Reserve Bank of St. Louis, OECD stands for the Organization for Economic Cooperation and Development, and FM means Freddie Mac. FHFA stands for Federal Housing Finance Agency.

B Markov switching models: estimation

For simplicity, let us assume a two-state Markov switching model such that states s_t can be equal to 1 or 2. Let θ be the vector of parameters entering the likelihood function for the data, and assume that the density conditional on being in state j, $\eta(HP_t | s_t = j, X_t, Y_t; \theta)$ is Gaussian:

$$\eta(HP_t \mid \Omega_t, s_t = j; \theta) = \frac{1}{\sqrt{2\pi h_j}} \exp\left(\frac{-(HP_t - \beta_{0,s_t} - \beta_1' X_t - \beta_{2,s_t}' Y_t)^2}{2h_j}\right)$$
(13)

for j = 1,2. The information set Ω_t contains HP_t , X_t , Y_t , ω_t and the lagged values of these variables, such that $\Omega_t = \{HP_t, X_t, Y_t, \omega_t, \Omega_{t-1}\}$.

Note that in this formulation, I assume a constant relationship between the conditioning factors Y_t and house prices within each state but allow these coefficients to vary between states. Alternatively, the relationship between conditioning factors X_t and house prices is constant.

The log-likelihood function takes the following form:

$$\ell(HP_t \mid \Omega_t; \theta) = \sum_{t=1}^T \ln(\phi(HP_t \mid \Omega_t; \theta))$$
(14)

where the density $\phi(HP_t \mid \Omega_t; \theta)$ is obtained by summing the weighted probability state densities across the two possible states such that:

$$\phi(HP_t \mid \Omega_t; \theta) = \sum_{j=1}^{2} \eta(HP_t \mid \Omega_t, s_t = j; \theta) P(s_t = j \mid \Omega_t; \theta)$$
(15)

where $P(s_t = j \mid \Omega_t; \theta)$ is the conditional probability of being in state *j* at time *t* given the information set Ω_t .

The conditional state probabilities can be obtained recursively such that:

$$P(s_{t} = i \mid \Omega_{t}; \theta) = \sum_{j=1}^{2} P(s_{t} = i \mid s_{t-1} = j, \Omega_{t}; \theta) P(s_{t-1} = j \mid \Omega_{t}; \theta)$$
(16)

Finally, using Bayes' rule, the conditional state probabilities can be written as:

$$P(s_{t-1} = j \mid \Omega_{t-1}; \theta) = P(s_{t-1} = j \mid HP_{t-1}, X_{t-1}, Y_{t-1}, \omega_{t-1}, \Omega_{t-2}; \theta)$$

=
$$\frac{\eta(HP_{t-1} \mid s_{t-1} = j, X_{t-1}, Y_{t-1}, \omega_{t-1}, \Omega_{t-2}; \theta) P(s_{t-1} = j \mid X_{t-1}, Y_{t-1}, \omega_{t-1}, \Omega_{t-2}; \theta)}{\sum_{j=1}^{2} \eta(HP_{t-1} \mid s_{t-1} = j, X_{t-1}, Y_{t-1}, \omega_{t-1}, \Omega_{t-2}; \theta) P(s_{t-1} = j \mid X_{t-1}, Y_{t-1}, \omega_{t-1}, \Omega_{t-2}; \theta)}$$
(17)

Markov switching models: Additional estimates С

	Mod	Model 1		el 2
Time-varying transition probab	oilities			
p(1,1), constant	1.911***	(0.31)	1.804***	(0.26)
p(1,1), mortgage debt (D)	36.549*	(18.99)	-0.325	(24.35)
p(1,2), constant	-2.232**	(0.98)	-5.427	(60977)
p(1,2), mortgage debt (D)	-47.661	(172.5)	5.021	(56074)
p(1,3), constant	-1.648***	(0.34)	-2.120***	(0.76)
p(1,3), mortgage debt (D)	-21.129	(20.63)	5.330	(38.47)
p(1,4), constant	-	-	-1.797***	(0.39)
p(1,4), mortgage debt (D)	-	-	5.592	(28.32)
p(2,1), constant	6.702	(199.10)	11.052	(50053)
p(2,1), mortgage debt (D)	2457.02	(74936.2)	2457.86	(81786)
p(2,2), constant	1.843***	(0.52)	4.329	(10849)
p(2,2), mortgage debt (D)	-9.045	(38.74)	-0.032	(85828)
p(2,3), constant	-2.503	(2.95)	-4.564	(13152)
p(2,3), mortgage debt (D)	-49.035	(218.30)	0.001	(14925)
p(2,4), constant	-	-	-0.191	(1.33)
p(2,4), mortgage debt (D)	-	-	23.038	(83.22)
p(3,1), constant	-	-	11.005	(40063)
p(3,1), mortgage debt (D)	-	-	2457.0	(75847)
p(3,2), constant	-	-	-1.486***	(0.35)
p(3,2), mortgage debt (D)	-	-	-2.588	(22.85)
p(3,3), constant	-	-	1.542***	(0.42)
p(3,3), mortgage debt (D)	-	-	21.774	(26.38)
p(3,4), constant	-	-	-3.889	(17908)
p(3,4), mortgage debt (D)	-	-	0.048	(14117)

Table 1 (cont.): Markov switching model estimates.

Notes: Standard deviations between brackets. Significance levels at 1%, 5% and 10% are represented by ***, **, * asterisks. Standard errors calculated using the first partial derivatives of the log likelihood, i.e. the outer product matrix. The time-varying transition probabilities are denoted p(i,j), means probability of transition from state i to state j.

D Markov switching models: Statistical tests

	Model 1	Model 2
Number of states	3	4
Unrestricted log likelihood value	1705.78	1798.16
A) Tests for number of states		
Log likelihood value with		
maximum number of states $s = 2$	1535.93	-
p-value	0.00	-
Log likelihood value with		
maximum number of states $s = 3$	-	1705.78
p-value	-	0.00
B) Tests for identical mean parameters		
Log likelihood value with		
$\beta_{0,s_t=1} = \beta_{0,s_t=2} = \beta_{0,s_t=3}$	1639.26	-
p-value	0.00	-
Log likelihood value with		
$\beta_{0,s_t=1} = \beta_{0,s_t=2} = \beta_{0,s_t=3} = \beta_{0,s_t=4}$	-	1378.28
p-value	-	0.00
Log likelihood value with		
$\beta_{4,s_t=1} = \beta_{4,s_t=2} = \beta_{4,s_t=3}$	1678.56	-
p-value	0.01	-
Log likelihood value with		
$\beta_{4,s_t=1} = \beta_{4,s_t=2} = \beta_{4,s_t=3} = \beta_{4,s_t=4}$	-	1695.42
p-value	-	0.00
C) Tests for identical variance parameters		
Log likelihood value with		
$\lambda_{0,s_t=1} = \lambda_{0,s_t=2} = \lambda_{0,s_t=3}$	1644.97	-
p-value	0.00	-
Log likelihood value with		
$\lambda_{0,s_t=1} = \lambda_{0,s_t=2} = \lambda_{0,s_t=3} = \lambda_{0,s_t=4}$	-	1769.15
p-value	-	0.01
D) Tests for identical probability parameters		
Log likelihood value with		
$\pi_{1,p}=\pi_{1,q}=\pi_{1,z}$	1535.70	-
p-value	0.00	-
Log likelihood value with		
$\pi_{1,p} = \pi_{1,q} = \pi_{1,z} = \pi_{1,r}$	-	1689.96
p-value	-	0.00
E) Tests of heteroskedasticity		
Goldfeld-Quandt test on mortgage debt (D)		
F-statistic	0.29	0.31
Critical value	1.26	1.27

Table 8: Statistical tests.

Notes: See Section 3 for a description of models 1 and 2. The Goldfeld-Quandt test orders the resulting residuals of each model by the variable mortgage debt (D), and divides the full sample in two, omitting six central observations. Then, it tests the null hypothesis of homoskedasticity.

E Markov switching Model 1 results

Table 9 shows the latest transition probabilities for Markov switching Model 1.

Table 9: Latest transition probabilities matrix, Model 1.

		State in T-1			
		1	2	3	
	1	0.96	0.02	0.06	
State in T	2	0.00	0.95	0.01	
	3	0.04	0.03	0.93	

Figure 4 shows a comparison of the filtered (red line) and smoothed (blue line) Markov switching probabilities of each of the three states according to Model 1.



Figure 4: Markov switching probabilities of being in state S_t , Model 1.

F Factor models of housing demand and supply

The structure of the dynamic factor models is common for both demand and supply specifications. I first assume that housing demand and supply are better proxied by several indicators than taking one of them. Second, I assume that the fundamentals included in each of the models are reasonable proxies, as is commonly used in the literature. Third, using the common factor implies that co-movements between multiple time series in each model arise from a single common factor.

Let y_t denote an $i \times 1$ vector of housing fundamentals in stationary form and standardized form. The dynamic common factor model of housing demand (or supply) yields:

$$y_t = \gamma c_t + e_t \tag{18}$$

where c_t is the common factor that follows an autoregressive structure of order two, such that:

$$c_t = \phi_1 c_{t-1} + \phi_2 c_{t-2} + w_t \tag{19}$$

where $w_t \sim iid N(0, \sigma_w^2)$, and the errors $e_{i,t}$ in e_t yield:

$$e_{i,t} = \psi_{i,1}e_{i,t-1} + \psi_{i,2}e_{i,t-2} + \varepsilon_{i,t}$$
(20)

where $\varepsilon_{i,t} \sim iid \operatorname{N}(0, \sigma_i^2)$.

The selected housing demand fundamentals in log-differences are the working age population (aged 15-64 years old), compensation of employees, non-farm employees, and the CPI of rents of primary residence, which are standard measures of housing demand commonly used in the literature¹³. Alternatively, the factor model of housing supply includes three variables in logs: new one-family houses sold, building permits, and housing starts, which are commonly used in the literature to track housing supply development (see Hilbers et al., 2008).

Both the housing demand and supply models are estimated using the maximum likelihood, and the systems are updated using the Kalman filter. After standardizing the common factor and applying the mean and standard deviation of log-differenced housing prices HP_t , I obtain the common factor of the fundamental variables in a housing prices-comparable fashion called f_t . Then, as deviations from house price growth, I obtain a measure of overvaluation, O_t , such that:

$$O_t = HP_t - f_t \tag{21}$$

Finally, this time series of overvaluation O_t is used to generate a binary indicator of overvaluation IO_t such that:

$$IO_t(O_t) = \begin{cases} 1 & \text{if } O_t > 0 \text{ and } HP_t > 0 \\ 0 & \text{otherwise} \end{cases}$$
(22)

¹³See Girouard et al., (2006) for a review of studies on housing prices and fundamentals in OECD countries.

Figure 5 plots the dynamic common factor of housing demand (blue line) compared with the S&P Case-Shiller home price index (black line). According to this approach, the longest period of overvaluation corresponds to 70 months from April 2000 to January 2006, that is, the years before the Great Recession. Other periods of overvaluation were August 1985 to August 1986, March 2012 to October 2013, August 2016 to January 2018, and discontinued in February 2017.



Figure 5: Common factor of housing demand and house prices.

Figure 6 shows the dynamic common factor of the housing supply (green line). According to this setup, there were two periods of overvaluation, from January 1984 to April 1989 and from March 2012 to December 2018, both with some monthly discontinuities.



Figure 6: Common factor of housing supply and house prices.

Notes: The housing supply proxies included are *new one-family houses sold*, *building permits*, and *housing starts*. The correlation between the supply factor and house prices is 53.9%.

References

- Acharya, V., Le, H. T. and H. S. Shin (2017), "Bank Capital and Dividend Externalities", The Review of Financial Studies, Vol. 30, No. 3 (March 2017), pp. 988-1018.
- [2] Adelino, M., A. Schoar and F. Severino (2018), "Dynamics of Housing Debt in the Recent Boom and Great Recession", National Bureau of Economic Research, in: NBER Macroeconomics Annual 2017, vol. 32, Chapter 3, pp. 265-311.
- [3] Aikman, D. (2021), "The objectives of macroprudential mortgage measures: an exploration", King's College London, Qatar Centre for Global Banking and Finance, Working paper n. 2021/2.
- [4] Ampudia, M., Duca, M. L., Farkas, M., Pérez-Quirós, G., Pirovano, M., Runstler, G. and Tereanu, E. (2021), "On the effectiveness of macroprudential policy", European Central Bank, Working Paper Series, n. 2559.
- [5] Arce, F., Bengui, J. and J Bianchi (2023), "Overborrowing, Underborrowing, and Macroprudential Policy", Federal Reserve Bank of Chicago, WP 2023-20, May 2023.
- [6] Bai, J. and S. Ng (2002), "Determining the number of factors in approximate factor models", Econometrica, 70(1), 191–221.
- [7] Bank for International Settlements (2010), "Guidance for national authorities operating the countercyclical capital buffer", Bank for International Settlements, Basel Committee on Banking Supervision.
- [8] Bank for International Settlements (2011), "Basel III: A global regulatory framework for more resilient banks and banking systems", Bank for International Settlements, Basel Committee on Banking Supervision.
- Bank for International Settlements (2019a), "Towards a sectoral application of the countercyclical capital buffer", Bank for International Settlements, Basel Committee on Banking Supervision, Working Paper 36.
- [10] Bank for International Settlements (2019b), "Guiding principles for the operationalisation of a sectoral countercyclical capital buffer", Bank for International Settlements, Basel Committee on Banking Supervision.
- [11] Barth, J. R., Joo, S., Kim, H., Lee, K. B., Maglic, S. and X. Shen (2020), "Forecasting Net Charge-Off Rates of Banks: A PLS Approach", in "Handbook of financial econometrics, mathematics, statistics, and machine learning", edited by C. F. Lee and J. C. Lee, World Scientific Book Chapters, chapter 63, pp. 2265-2301, World Scientific Publishing Co. Pte. Ltd.
- [12] Bernanke B. (2002), "Asset-price "bubbles" and monetary policy", Remarks before the New York Chapter of the National Association for Business Economics, New York, October 15, 2002.
- [13] Borio, C., Farag, M. and F. Zampolli (2023), "Tackling the fiscal policy-financial stability nexus", BIS Working Papers, No 1090, April 2023.
- [14] Cerutti, E., Claessens, S. and L. Laeven (2017a), "The use and effectiveness of macroprudential policies: New evidence", Journal of Financial Stability, vol. 28, pp. 203–224.
- [15] Cerutti, E., Dagher, J. and G. Dell'Ariccia (2017b), "Housing finance and real-estate booms: A cross-country perspective", Journal of Housing Economics, vol. 38, pp. 1-13.
- [16] Coulter, J., Grossman, V., Martínez-García, E., Phillips, P. C. B. and S. Shi (2022), "Real-Time Market Monitoring Finds Signs of Brewing U.S. Housing Bubble", Federal Reserve Bank of Dallas.
- [17] Chow, G. and A. Lin (1971), "Best Linear Unbiased Interpolation, Distribution, and Extrapolation of Time Series by Related Series", The Review of Economics and Statistics, 1971, vol. 53, issue 4, pp. 372-75.
- [18] Crowe, C., Dell'Ariccia, G., Igan, D. and P. Rabanal (2013), "How to deal with real estate booms: Lessons from country experiences", Journal of Financial Stability, vol. 9, Issue 3, pp. 300-319.
- [19] Dautovic, E., Gambacorta, E. and A. Reghezza (2023), "Supervisory policy stimulus: evidence from the euro area dividend recommendation", Bank for International Settlements, Monetary and Economic Department, BIS Working Papers No 1085.
- [20] Di Maggio, M. and A. Kermani (2017), "Credit-Induced Boom and Bust", The Review of Financial Studies, vol. 30, issue 11, pp. 3711–3758.
- [21] **Ding, Z.** (2023), "*Regime Switching Model with Time Varying Transition Probabilities*", MATLAB Central File Exchange (https://www.mathworks.com/matlabcentral/fileexchange/37144-regime-switching-model-with-time-varying-transition-probabilities).
- [22] Döme, S. and M. Sigmund (2023), "How do macroprudential authorities set countercyclical capital buffers?", Oesterreichische Nationalbank, manuscript.
- [23] Drehmann, M., Borio, C., Gambacorta, L., Jiménez, G. and C. Trucharte (2010), "Countercyclical capital buffers: Exploring options", Bank for International Settlements, Working Papers, n. 317.
- [24] Duca, J., Muellbauer, J. and A. Murphy (2010), "Housing markets and the financial crisis of 2007–2009: Lessons for the future", Journal of Financial Stability, vol. 6, issue 4, pp. 203–217.
- [25] European Systemic Risk Board (2022), "Vulnerabilities in the residential real estate sectors of the EEA countries", February 2022.
- [26] Fabozzi, F. J., Kynigakis, I., Panopoulou, E. and R. Tunaru (2020), "Detecting Bubbles in the US and UK Real Estate Markets", The Journal of Real Estate Finance and Economics, vol. 60, pp. 469–513.
- [27] Favara, G. and J. Imbs (2015), "Credit Supply and the Price of Housing", American Economic Review, vol. 105, n. 3, pp. 958–992.

- [28] Favilukis, J., S. Ludvigson, and S. V. Nieuwerburgh (2017), "The macroeconomic effects of housing wealth, housing finance, and limited risksharing in general equilibrium", Journal of Political Economy, vol. 125, n. 1, February 2017.
- [29] Gatt, W. (2023), "Housing Boom-Bust Cycles and Asymmetric Macroprudential Policy", Journal of Money, Credit and Banking, August 2023.
- [30] Gervais, M. (2002), "Housing taxation and capital accumulation", Journal of Monetary Economics, Vol. 49, Issue 7, October 2002, pp. 1461-1489.
- [31] Girouard, N., Kennedy, M., Van den Noord, P. and C. André (2006), "Recent House Price Developments: The Role of Fundamentals", OECD, Economics Department Working Papers, n. 475.
- [32] Greenwald, D. (2018), "The Mortgage Credit Channel of Macroeconomic Transmission", MIT Sloan Research Paper, n. 5184-16.
- [33] Greenwald, D. and A. M. Guren (2021), "Do Credit Conditions Move House Prices?", NBER Working Paper, n. w29391.
- [34] Greenwood, R., Hanson, S., Shleifer, A. and J. A. Sørensen (2022), "Predictable Financial Crises", Journal of Finance, vol. 77, Issue 2, April 2022, pp. 863-921.
- [35] Guerrieri, L. and M. Iacoviello (2017), "Collateral Constraints and Macroeconomic Symmetries", Journal of Monetary Economics, 90, pp. 28–49.
- [36] Guerrieri, V. and H. Uhlig (2016), "Housing and Credit Markets", in "Handbook of Macroeconomics", Chapter Chapter 17, 2016, vol. 2, pp 1427-1496, Elsevier.
- [37] Gürkaynak, R. (2008), "Econometric tests of asset price bubbles: taking stock", Journal of Economic Surveys, vol. 22, n. 1, pp. 166–186.
- [38] Hamilton, J. D. (1989), "A New Approach to the Economic Analysis of Nonstationary Time Series and the Business Cycle", Econometrica, Vol. 57, No. 2 (Mar., 1989), pp. 357-384.
- [39] Hardy, B. (2021), "Covid-19 bank dividend payout restrictions. Effects and trade-offs", BIS Bulletin, no. 38.
- [40] Hilbers, P., Hoffmaister, A., Angana Banerji, A. and H. Shi (2008), "House Price Developments in Europe: A Comparison", IMF, Working Paper n. 08/211.
- [41] Iacoviello, M. (2005), "House Prices, Borrowing Constraints, and Monetary Policy in the Business Cycle", American Economic Review, n. 95, pp. 739–764.
- [42] International Monetary Fund (2019), "Downside risks to house prices", Global Financial Stability Report, April 2019, Chapter 2.
- [43] Joint Center for Housing Studies of Harvard University (2018), "The State of the Nation's Housing 2018", Harvard University.
- [44] Jordà, Ò., Schularick, M. and A. M. Taylor (2013), "When Credit Bites Back", Journal of Money, Credit and Banking, Vol. 45, Supplement 2 (December 2013), pp. 3-28.
- [45] Jordà, Ò., Schularick, M. and A. M. Taylor (2015a), "Leveraged Bubbles", Journal of Monetary Economics, 76, S1–S20.
- [46] Jordà, Ò., Schularick, M. and A. M. Taylor (2015b), "Betting the house", Journal of International Economics, 96, S2–S18.
- [47] Jordà, Ò., Schularick, M. and A. M. Taylor (2016), "The great mortgaging: housing finance, crises and business cycles", Economic Policy, vol. 31, Issue 85, pp. 107–152.
- [48] Jordà, Ò., Schularick, M. and A. M. Taylor (2017), "Macrofinancial History and the New Business Cycle Facts", National Bureau of Economic Research, Macroeconomics Annual 2016, vol. 31.
- [49] Justiniano, A., Primiceri, G. E. and A. Tambalotti (2019), "Credit Supply and the Housing Boom", Journal of Political Economy, vol. 127, n. 3, June 2019.
- [50] Karlman, M., Kinnerud, K. and K. Kragh-Sørensen (2021), "Costly reversals of bad policies: The case of the mortgage interest deduction", Review of Economic Dynamics, Volume 40, April 2021, pp. 85-107.
- [51] Kim, H. and J. Son (2023), "What Charge-Off Rates Are Predictable by Macroeconomic Latent Factors?", Munich Personal RePEc Archive (MPRA), Paper No. 116880
- [52] Kiyotaki N. and J. Moore (1997), "Credit cycles", Journal of Political Economy, vol. 105, n. 2, pp. 211-248.
- [53] Mendicino, C., Nikolov, K., Suárez, J. and D. Supera (2020), "Bank capital in the short and in the long run", Journal of Monetary Economics, vol. 115, November 2020, pp. 64-79.
- [54] Mian, A. and A. Sufi (2009), "The Consequences of Mortgage Credit Expansion: Evidence from the U.S. Mortgage Default Crisis", Quarterly Journal of Economics, vol. 124, n.4, pp. 1449–1496.
- [55] Mian, A. and A. Sufi (2010), "Household Leverage and the Recession of 2007 to 2009", IMF Economic Review, vol. 58, n. 1, pp. 74-117.
- [56] Mian, A. and A. Sufi (2011), "House Prices, Home Equity-Based Borrowing, and the US Household Leverage Crisis", American Economic Review, vol. 101, n. 5, pp. 2132–2156.
- [57] Mian, A. and A. Sufi (2018), "Finance and Business Cycles: The Credit-Driven Household Demand Channel", Journal of Economic Perspectives, vol. 32, n. 3, Summer 2018, pp. 31–58.
- [58] Muellbauer, J. (2012), "When is a Housing Market Overheated Enough to Threaten Stability?", Reserve Bank of Australia, RBA Annual Conference Volume 2012, in: Property Markets and Financial Stability, pp. 73-105.

- [59] Muñoz, M. A. (2021), "Rethinking Capital Regulation: The Case for a Dividend Prudential Target", International Journal of Central Banking, vol. 17(3), pp. 271-336.
- [60] Müller, K. (2022), "Electoral Cycles in Macroprudential Regulation", American Economic Journal: Economic Policy (forthcoming).
- [61] Nneji, O., Brooks, C. Ward, C. (2013), "House Price Dynamics and Their Reaction to Macroeconomic Changes", Economic Modelling, vol. 32, pp. 172-178.
- [62] Pavlidis, E., Yusupova, A., Paya, I., Peel, D., Mack, A. and V. Grossman (2016), "Episodes of Exuberance in Housing Markets: In Search of the Smoking Gun", Journal of Real Estate Finance and Economics, vol. 53, issue 4, pp. 419–449.
- [63] Phillips, P. C. B., Shi, S. and J. Yu (2015), "Testing for multiple bubbles: Historical episodes of exuberance and collapse in the S&P 500.", Cowles Foundation, Discussion Paper n. 1498.
- [64] Phillips, P. C. B., Wu, Y. and J. Yu (2011), "Explosive Behavior in the 1990s Nasdaq: When Did Exuberance Escalate Asset Values?", International Economic Review, vol. 52, issue 1, pp. 201-226.
- [65] Rappaport, J. (2016), "The Limited Supply of Homes", Federal Reserve Bank of Kansas City, The Macro Bulletin, March 23, 2016.
- [66] Roubini, N. (2006), "Why Central Banks Should Burst Bubbles", International Finance, vol. 9, issue 1, pp. 87-107.
- [67] Schularick, M. and A. M. Taylor (2012), "Credit Booms Gone Bust: Monetary Policy, Leverage Cycles, and Financial Crises, 1870-2008", American Economic Review, vol. 102, n. 2, pp. 1029-1061.
- [68] Sheng, X., Gupta, R. and Q. Ji (2022), "Forecasting charge-off rates with a panel Tobit model: the role of uncertainty", Applied Economics Letters, vol. 29, Issue 10.
- [69] Shi, S. (2017), "Speculative bubbles or market fundamentals? An investigation of US regional housing markets", Economic Modelling, vol. 66, pp. 101–111.
- [70] Standard & Poor's Dow Jones Indices (2019), "S&P CoreLogic Case-Shiller Home Price Indices Methodology", https://us.spindices.com
- [71] Stein, J. C. (2021), "Can policy tame the credit cycle?", IMF, Economic Review, 69, pp. 5–22.
- [72] Stiglitz, J. E. (1990), "Symposium on Bubbles", Journal of Economic Perspectives, Spring, 4:2, pp. 13–18.
- [73] Van Norden, S. and H. Schaller (1993), "The predictability of stock market regime: Evidence from the Toronto stock exchange", The Review of Economics and Statistics, 75(3), pp. 505–510.
- [74] Van Norden, S. and H. Schaller (1996), "Speculative behaviour, regime-switching, and stock market crashes", Bank of Canada, Working Paper 96-13.
- [75] Whitehouse, E. J, Harvey, D. I. and S. J. Leybourne (2023), "Real-Time Monitoring of Bubbles and Crashes", Oxford Bulletin of Economics and Statistics, vol. 85, Issue 3, Jun 2023, pp. i-iii, 457-670.